Author’s response to reviews

Title: Comparison of outcomes between emergent-start and planned-start peritoneal dialysis in incident ESRD patients: a prospective observational study

Authors:
Wen-Yi Li (b5401068@gmail.com)
Yi-Cheng Wang (birdhead0729@gmail.com)
Shang-Jyh Hwang (sjhwang@kmu.edu.tw)
Shih-Hua Lin (l521116@ndmctsgh.edu.tw)
Kwan-Dun Wu (kdwu@ntu.edu.tw)
Yung-Ming Chen (chenym055@gmail.com)

Version: 2 Date: 19 Nov 2017

Author’s response to reviews:
Dear Editor:

Thank you very much for giving us an opportunity to revise the manuscript. According to your suggestions, we have revised the manuscript and submit it as a single, final, clean version. Our point-by-point responses are detailed below.

1. All figures are uploaded as separate files.
2. Institutional email addresses for all authors are provided on the title page.
3. The original funding information has been relocated to the Funding section, and “Not applicable” has been put under the Acknowledgements section. (P.13-14.)
4. The heading has been changed to Methods. (P.5, L.1)
5. Additionally, we have updated the word count of the main text (now 2683), and corrected some typos in the References section.

I hope the revised version meet the high standard of BMC Nephrology. If you have any further queries, please contact us at your earliest convenience.
Sincerely yours,

Wen-Yi Li, MD

Renal Division, Department of Internal Medicine, National Taiwan University
Hospital Yunlin Branch, No.579, Sec. 2, Yunlin Rd., Douliu City, Yunlin County 640, Taiwan
Telephone: +886-5-5323911 Ext 6203
Fax: +886-5-5374783
E-mail: b5401068@gmail.com